Author’s response to reviews

Title: Confirming mental health care in acute psychiatric wards, as narrated by persons experiencing psychotic illness: an interview study

Authors:

Karina Sebergsen (karina.sebergsen@unn.no)
Astrid Norberg (astrid.norberg@umu.se)
Anne-Grethe Talseth (anne.g.talseth@uit.no)

Version: 2 Date: 17 Dec 2015

Author’s response to reviews:

Response to the editorial request:

1. We removed the funding section to the acknowledgements section.

2. A separate figure legend for Figure 1 is at the end of the manuscript.

3. We ensure that our manuscript adheres to COREQ guidelines.

To note, related the COREQ guidelines, we now inform in the chapter of Recruitment procedure in the main manuscript the number of requested persons to participate in our study.